Author's response to reviews

Title: Accelerated tibial fracture union in the third trimester of pregnancy; a case report.

Authors:

Mudussar A Ahmad (mudussarahmad@hotmail.com)
Damayanthi Kuhanendran (damy007@hotmail.com)
Irvine W Kamande (Irvinekamande@hotmail.com)
Charalambos Charalambides
(charalambos.charalambides@whittington.nhs.uk)

Version: 7 Date: 1 February 2008

Author's response to reviews:

Formatting changes -------------------------------------

Title - Please ensure the manuscript title is identical in both the manuscript and submission system. DONE

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. DONE

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Please link each author to his or her corresponding affiliation with the use of superscript numbers. DONE

Author email addresses - The corresponding author email address must be given on the title page. If possible please give a different email address for each author, on the title page, beneath the corresponding author details. Please link authors to their email addresses using their initials, for example:

ABC: Myemail@address.com DONE

Abstract - Please move the abstract down to the next page. The abstract must be structured into the following sections:

Introduction
Case presentation
Conclusions. DONE

Please ensure the abstract is identical in both the manuscript and submission system. DONE
Please use sentence case for all headings, such as change Figures Legends to Figure legends (lower case l). DONE

References - References should be cited in the text using consecutive numbers in square brackets starting from [1] then [2] etc. Please ensure the type of bracket used is consistent. Please do not use superscript numbers. DONE

Case report - Please rename this section to Case presentation. DONE

Consent - Please reposition the consent section after the Authors' contributions. Please expand the statement of consent to include the figures, for example: "Written informed consent was obtained from the patient for publication of this Case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal." DONE

Competing interests - Please include a competing interests section above the Authors' contributions. Please place this statement The authors declare that they have no competing interests. within the competing interests section. DONE.

Acknowledgements - Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Place the section directly before the references in the main body of the text. DONE.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements. DONE.

Please use sentence case for all headings, such as change Figures Legends to Figure legends (lower case l). DONE.